Author's response to reviews

Title: Why not just Google it? An assessment of information literacy skills in a biomedical science curriculum.

Authors:

Karl Kingsley (karl.kingsley@unlv.edu)
Gillian M Galbraith (gillian.galbraith@unlv.edu)
Matthew Herring (matthew.herring@sdm.unlv.edu)
Eva Stowers (eva.stowers@unlv.edu)
Tanis Stewart (tastewart@fdic.gov)
Karla V Kingsley (karlak@unm.edu)

Version: 5 Date: 12 April 2011

Author's response to reviews:

BMC Editorial Team:

Per your request dated April 10, 2011, the following formatting changes have been made:

1. Tables have been made in the MS Word auto-format template: Simple 1 with no vertical lines;
2. Line numbers have been removed;
3. Figure 1 has been recropped to remove excess space;

Many thanks for your assistance,

Karl Kingsley